\begin{table}%tB.1%tB.2
The numerical information associated with Fig.~\ref{SpecVar20052007_mix} is given in Tables~\ref{SpecVarFTabLeftGrey} (left panel, grey points), \ref{SpecVarFTabLeftBlack} (left panel, black points) and~\ref{SpecVarFTabRight} (right panel). In~addition, numerical information associated with Fig.~\ref{FourNightsLC} is given in Table~\ref{SpecVar4nightsTab}.
\par
\vspace*{2mm}
\setcounter{table}{0}
\renewcommand{\thetable}{B.\arabic{table}}
\caption{\label{SpecVarFTabLeftGrey}Integral flux ($10^{-11}~\percmsqrs$) in the 0.2$-$1~TeV energy range versus photon index corresponding to grey points in the left panel of Fig.~\ref{SpecVar20052007_mix}.} \vspace*{-3mm}
\par
%\centering
\par
 { {
\begin{tabular}{*{2}{c}}
\hline \hline \noalign{\smallskip}
$\Phi$ & Index $\Gamma$ \\ 
   \hline
$2.36\pm0.13$ & $3.345\pm0.20$ \\
$3.92\pm0.17$ & $3.64\pm0.16$ \\
$5.33\pm0.22$ & $3.46\pm0.13$ \\
$8.29\pm0.30$ & $3.64\pm0.10$ \\
$13.82\pm0.82$ & $3.82\pm0.17$ \\
   \hline
\end{tabular}}} \vspace*{-2mm}
\tablefoot {Errors are statistical. See Sect.~\ref{SpecVarAll} for more details.}
\end{table}